CS150 Spring 97

Page 1.1. 1

CS150 Newton 15.2. 1

Outline

m Last time:

Ô Error Detection & Correction

Ô State Assignment in Asynchronous

m This lecture:

Ô Review

Ô Analog-Digital Conversion

Ô Counters and analysis

CS150 Newton 15.2. 2

Analog-to-Digital Conversion:

Conversion to Binary

+

-

C

I

Vin

Binary Counter

Count pulses

Simple slope conversion

approach

CS150 Spring 97

Page 1.1. 2

CS150 Newton 15.2. 3 m Obtain the state table for the circuit below, starting from

the state Q 3 Q 2 Q 1 =000.

CS150 Newton 15.2. 4

Example 1

m State table:

CS150 Spring 97

Page 1.1. 3

CS150 Newton 15.2. 5 m Obtain a synchronous counter that produces the count

sequence 0, 2, 4, 3, 6, 7, 0, ...

CS150 Newton 15.2. 6

Example 2

m J & K Boolean equations:

CS150 Spring 97

Page 1.1. 4

CS150 Newton 15.2. 7 m Describe the function and characteristics of the circuit

shown below: